Author's response to reviews

Title: Leontiasis ossea and post traumatic cervical cord contusion in polyostotic fibrous dysplasia

Authors:

Boby V Maramattom (bob.varkey@yahoo.com)

Version: 2 Date: 6 June 2006

Author's response to reviews:

As required by reviewer 2, more x-rays have been added as well as T1 weighted MRI's. Other causes of cervical cord injury, specifically dislocations, fractures etc have been and the extent of cervical dysplasia is clearly visible on the cervical X-rays and MRI images. The minor inconsistencies regarding the type of flexion/extension injury have also been corrected. More detail has been added regarding the involvement of the spine in PFD and the importance of cervical canal stenosis in PFD.